Dear Editor

I hope everything is all right.

Thank you very much for your email of July 15, 2021, informing us of tentative acceptance of the above referenced paper. We thank everyone for their hard work and help during the COVID-19 pandemic on us all. Attached please find the revised manuscript. Our point-by-point responses to the reviewers’ comments are as follows.

Responses to Referee 1’s critiques “The study is aimed at establishment of the MM spine stability scoring system, which can serve to provide a vital theoretical basis for the evaluation of spine stability in individuals with MM. The idea is novel and justified. The methodology based on DELPHI is robust and covers vital prerequisites. Moreover, the application on real time cases has been deemed appropriate. The limitations and future implications have been adequately addressed as well.”

Reply: Thanks reviewer for good comments and hard work.